Ondansetron for irritable bowel syndrome with diarrhoea: randomised controlled trial

David Gunn,^{1,2} Rabia Topan,³ Ron Fried,⁴ Ivana Holloway,⁵ Richard Brindle,⁵ Suzanne Hartley,⁵ Lorna Barnard,⁵ Maura Corsetti,² S Mark Scott,³ Adam Farmer,⁶ Ayesha Akbar,⁷ Maria Eugenicos,⁸ Nigel Trudgill,⁹ Kapil Kapur,¹⁰ John McLaughlin,¹¹ David S Sanders,¹² Arvind Ramadas,¹³ Peter Whorwell,¹⁴ Lesley Houghton,¹⁵ Phil G Dinning,¹⁶ Qasim Aziz,³ Alexander C Ford,^{17,18} Amanda Farrin⁵ and Robin Spiller^{1,2*}

- ¹NIHR Nottingham Digestive Diseases Biomedical Research Centre, University of Nottingham, Nottingham, UK
- ²Nottingham Digestive Diseases Centre, Nottingham University Hospitals NHS Trust, Nottingham, UK
- ³Barts and the London School of Medicine and Dentistry, Queen Mary University of London, London, UK
- ⁴Department of Gastroenterology and Hepatology, University of Zurich, Zurich, Switzerland
- ⁵Clinical Trials Research Unit, Leeds Institute of Clinical Trials Research, University of Leeds, Leeds, UK
- ⁶Royal Stoke Hospital, University Hospitals of North Midlands NHS Trust, Stoke, UK ⁷St Mark's Hospital, Harrow, UK
- ⁸University of Edinburgh, Western General Hospital, Edinburgh, UK
- ⁹Sandwell and West Birmingham NHS Trust, Birmingham, UK
- ¹⁰Barnsley Hospital NHS Foundation Trust, Barnsley, UK
- ¹¹University NHS Foundation Trust, Manchester, UK
- ¹²Royal Hallamshire Hospital, Sheffield Teaching Hospitals NHS Foundation Trust, Sheffield Wythenshawe Hospital, Manchester, UK
- ¹³Tees Valley Hospital, Middlesbrough, UK
- ¹⁴University of Manchester, Manchester, UK
- ¹⁵St James's University Hospital, Leeds, UK
- ¹⁶Discipline of Surgery and Gastroenterology, Flinders Medical Centre, Flinders University, South Australia, Australia
- ¹⁷Leeds Institute of Medical Research at St. James's, University of Leeds, Leeds, UK
- ¹⁸Leeds Gastroenterology Institute, Leeds Teaching Hospitals Trust, Leeds, UK

*Corresponding author robin.spiller@nottingham.ac.uk

Disclosure of interests

Full disclosure of interests: Completed ICMJE forms for all authors, including all related interests, are available in the toolkit on the NIHR Journals Library report publication page at https://doi.org/10.3310/YTFW7874.

Primary conflicts of interest: Robin Spiller has received research grants from Zespri International and Sanofi-Aventis and speakers'f fees from Ardelyx, Menarini and Ferrer. S Mark Scott has received payment or honoraria for lectures, presentations, speakers bureaus, manuscript writing or educational events Medical Measurement Systems (MMS)/Laborie. John McLaughlin is an unpaid trustee and deputy chair, Guts UK charity. Ayesha Akbar has received speakers' fees from the following pharmaceutical companies: Takeda, Dr Falk, Janssen, Tillots and Galapagos. Maura Corsetti is Co-Chief Investigator in a research grant funded by Sanofi, has received consulting fees from Sanofi, Takeda, Mayoly, RB and Arena, and has received payments or honoraria for lectures, presentations, speakers bureaus, manuscript writing or educational events from Sanofi and Mayoly. Lesley Houghton has received consulting fees from Ironwood Pharmaceuticals, USA, and has a leadership or fiduciary role in other boards, societies, committees or advocacy groups, paid or unpaid in Rome V International Working Group - Fundamentals of Neurogastroenterology: Physiological Aspects and Clinical Implications Section Chair, Member of British Society of Gastroenterology Committee – Guidelines for irritable bowel syndrome, and Member of British Society of Gastroenterology Committee – Guidelines for functional dyspepsia. Qasim Aziz has received grants or contracts from any entity from Classado Pharmaceuticals and Allergan, has received payment or honoraria for lectures, presentations, speakers bureaus, manuscript writing or educational events from Bromatec, has participated on a Data Safety Monitoring Board or Advisory Board for the MHRA GRID EAG Committee, has a leadership or fiduciary role in other boards, societies, committees or advocacy groups, paid or unpaid at the European Society of Neurogastroenterology and Motility, and has stock or stock options in My Health Chart, The Physicians' Clinic and London Digestive Centre. Amanda Farrin is a NIHR Senior Investigator, has received research grants from NIHR and was a member of HTA Efficient Study Designs Board (13 October 2014–17 December 2014), NIHR CTU Standing Advisory Committee (1 February 2022–1 May 2026), HTA Funding Committee Policy Group (formerly CSG) (12 March 2014-31 October 2018) and the HTA Clinical Evaluation and Trials Committee (1 November 2014–30 November 2018). Suzanne Hartley has support for the present manuscript, role on the project was partly funded by the project grant (National Institute for Health Research Efficacy and Mechanism Evaluation) and grants or contracts from any entity, role has been partly funded by other NIHR grants. Alexander C Ford has received consulting and/or speakers' fees from Dr Falk, GE Healthcare, Ironwood, Novartis and Takeda Pharmaceuticals, and has received research funding from Dr Falk and Tillotts, UK.

Published October 2023 DOI: 10.3310/YTFW7874

Plain language summary

Ondansetron for irritable bowel syndrome with diarrhoea: randomised controlled trial Efficacy and Mechanism Evaluation 2023; Vol. 10: No. 9 DOI: 10.3310/YTFW7874

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain language summary

Background

Irritable bowel syndrome with diarrhoea is characterised by frequent, loose, or watery bowel movements with marked reduction of quality of life. A previous small study suggested ondansetron benefits patients with irritable bowel syndrome with diarrhoea.

Methods

A clinical trial aiming to recruit 400 patients meeting established criteria for irritable bowel syndrome with diarrhoea from 18 centres throughout the UK. Patients received either ondansetron or placebo for 12 weeks but neither the investigator and nor patient could tell which they were receiving. They recorded their worst abdominal pain, stool frequency and consistency daily. The main end point was the proportion of patients meeting a standard recommended by the U.S. Food and Drug Administration (FDA). Being called a "FDA responder" meant they showed reductions to both pain and days with loose bowel movements. Other less important end points included pain intensity, stool consistency and frequency. We also measured the time for content to pass through the gut (whole gut transit time).

Results

The study closed early due to slow recruitment with 80 patients randomised. There were 40.5% of responders in the ondansetron group and 27.9% in the placebo group; however, due to low numbers these differences could be due to chance. Ondansetron produced a significant improvement in average stool consistency in the final month of treatment. Ondansetron slowed whole gut transit time which increased from baseline to 12 weeks by a mean of 3.8 hours, while it fell 2.2 hours on placebo, a difference unlikely to be due to chance.

Conclusion

These results are consistent with previous studies showing that ondansetron improves stool consistency and slows transit. However, because the numbers recruited were smaller than planned, the apparent improvement in "FDA responder" rate could have been due to chance. A further larger trial is needed to confirm the benefit of ondansetron which should be done in primary care where most patients are to be found.

Efficacy and Mechanism Evaluation

ISSN 2050-4365 (Print)

ISSN 2050-4373 (Online)

Efficacy and Mechanism Evaluation (EME) was launched in 2014 and is indexed by Europe PMC, DOAJ, Ulrichsweb[™] (ProQuest LLC, Ann Arbor, MI, USA) and NCBI Bookshelf.

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full EME archive is freely available to view online at www.journalslibrary.nihr.ac.uk/eme

Criteria for inclusion in the Efficacy and Mechanism Evaluation journal

Reports are published in *Efficacy and Mechanism Evaluation* (EME) if (1) they have resulted from work for the EME programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

EME programme

The Efficacy and Mechanism Evaluation (EME) programme funds ambitious studies evaluating interventions that have the potential to make a step-change in the promotion of health, treatment of disease and improvement of rehabilitation or long-term care. Within these studies, EME supports research to improve the understanding of the mechanisms of both diseases and treatments.

The programme supports translational research into a wide range of new or repurposed interventions. These may include diagnostic or prognostic tests and decision-making tools, therapeutics or psychological treatments, medical devices, and public health initiatives delivered in the NHS.

The EME programme supports clinical trials and studies with other robust designs, which test the efficacy of interventions, and which may use clinical or well-validated surrogate outcomes. It only supports studies in man and where there is adequate proof of concept. The programme encourages hypothesis-driven mechanistic studies, integrated within the efficacy study, that explore the mechanisms of action of the intervention or the disease, the cause of differing responses, or improve the understanding of adverse effects. It funds similar mechanistic studies linked to studies funded by any NIHR programme.

The EME programme is funded by the Medical Research Council (MRC) and the National Institute for Health and Care Research (NIHR), with contributions from the Chief Scientist Office (CSO) in Scotland and National Institute for Social Care and Health Research (NISCHR) in Wales and the Health and Social Care Research and Development (HSC R&D), Public Health Agency in Northern Ireland.

This report

The research reported in this issue of the journal was funded by the EME programme as project number 15/74/01. The contractual start date was in February 2017. The final report began editorial review in May 2022 and was accepted for publication in February 2023. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The EME editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research. The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, the MRC, the EME programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, the EME programme or the Department of Health and Social Care.

Copyright © 2023 Gunn *et al.* This work was produced by Gunn *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaptation in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Newgen Digitalworks Pvt Ltd, Chennai, India (www.newgen.co).

NIHR Journals Library Editor-in-Chief

Dr Cat Chatfield Director of Health Services Research UK

NIHR Journals Library Editors

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HSDR, PGfAR, PHR journals) and Editorin-Chief of HSDR, PGfAR, PHR journals

Dr Peter Davidson Interim Chair of HTA and EME Editorial Board, Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Consultant in Public Health, Delta Public Health Consulting Ltd, UK

Ms Tara Lamont Senior Adviser, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Dr Catriona McDaid Reader in Trials, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor James Raftery Professor of Health Technology Assessment, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Dr Rob Riemsma Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Professor Helen Roberts Professor of Child Health Research, Child and Adolescent Mental Health, Palliative Care and Paediatrics Unit, Population Policy and Practice Programme, UCL Great Ormond Street Institute of Child Health, London, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk